Author's response to reviews

Title: Burden of morbidity in an individual perspective - the case of sick-listed patients in primary care

Authors:

Lennart Carlsson (lennart.carlsson@ki.se)
Roland Morgell (roland.morgell@ki.se)
Lars-Erik Stender (lars-erik.stender@ki.se)
Britt Arrelöv (britt.arrelov@sll.se)
Gunnar H Nilsson (gunnar.nilsson@ki.se)

Version: 2 Date: 16 June 2008

Author's response to reviews:

Amendments asked for by you have been included;
Background in the abstract - declaration of interests - Authors contribution - references.